
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Mon Sep 15 16:47:01 GMT 1997
Year: 97  Doy: 258
Observer: yasukawa
Mon Sep 15 16:47:04 GMT 1997    CHIP     Startup--Initializing new tape
WEATHER COMMENT: Mon Sep 15 16:48:18 GMT 1997
Cold, in blowing mist, brisk south wind.  Waiting for weather to improve 
before opening up dome.
LOW-L COMMENT: Mon Sep 15 17:01:25 GMT 1997
File number was 71 yesterday and 7 this morning on drive 0.
Popped L00472 out and cleaned drive, then loaded L00472 into drive 0,
and L00474 into drive 1.  Cleaned window.
WEATHER COMMENT: Mon Sep 15 18:38:22 GMT 1997
Cap cloud moved off, still in broken cirrus but able to start CHIP and DPMON.
Mon Sep 15 18:39:19 GMT 1997    dPMon    Start Patrol
Mon Sep 15 18:39:44 GMT 1997    CHIP     CHIP Start Patrol
Mon Sep 15 19:01:04 GMT 1997    dPMon    Flat
Mon Sep 15 19:01:12 GMT 1997    CHIP     Bias
Mon Sep 15 19:02:00 GMT 1997    dPMon    End Flat
Mon Sep 15 19:02:08 GMT 1997    CHIP     End Bias
Mon Sep 15 19:02:18 GMT 1997    CHIP     Water
Mon Sep 15 19:02:50 GMT 1997    CHIP     End Water
Mon Sep 15 19:22:39 GMT 1997    MkIII    Start Patrol
Mon Sep 15 19:50:15 GMT 1997    MkIII    End Patrol
Mon Sep 15 19:50:37 GMT 1997    MkIII    Start Patrol
Mon Sep 15 19:58:32 GMT 1997 
        MkIII   
 
97d258.18_54.rawmk3  97d258.19_14.rawmk3  97d258.19_50.rawmk3
97d258.18_57.rawmk3  97d258.19_22.rawmk3  97d258.19_54.rawmk3
Mon Sep 15 20:00:58 GMT 1997    dPMon    Flat
Mon Sep 15 20:01:02 GMT 1997    CHIP     Gain
Mon Sep 15 20:01:27 GMT 1997 
        MkIII   
 
97d258.19_57.rawmk3
Mon Sep 15 20:01:56 GMT 1997    dPMon    End Flat
Mon Sep 15 20:05:16 GMT 1997    CHIP     End Gain
Mon Sep 15 20:05:26 GMT 1997    CHIP     Bias
Mon Sep 15 20:06:10 GMT 1997    CHIP     End Bias
Mon Sep 15 20:06:19 GMT 1997    CHIP     Water
Mon Sep 15 20:06:48 GMT 1997    CHIP     End Water
Mon Sep 15 21:00:59 GMT 1997    dPMon    Flat
Mon Sep 15 21:01:51 GMT 1997    dPMon    End Flat
Mon Sep 15 21:02:02 GMT 1997    CHIP     Bias
Mon Sep 15 21:02:49 GMT 1997    CHIP     End Bias



Mon Sep 15 21:03:00 GMT 1997    CHIP     Water
Mon Sep 15 21:03:33 GMT 1997    CHIP     End Water
Mon Sep 15 22:01:03 GMT 1997    CHIP     Bias
Mon Sep 15 22:01:50 GMT 1997    CHIP     End Bias
Mon Sep 15 22:01:57 GMT 1997    CHIP     Water
Mon Sep 15 22:02:28 GMT 1997    CHIP     End Water
Mon Sep 15 22:05:53 GMT 1997    MkIII    Start Patrol
DPMON COMMENT: Mon Sep 15 22:12:10 GMT 1997
DPMON crashed at some point.  We were working on MKIII modifications
and did not notice when this occurred.  Shutting it down.
Mon Sep 15 22:17:36 GMT 1997    CHIP     CHIP End Patrol
Mon Sep 15 22:18:01 GMT 1997    CHIP     ending tape
Mon Sep 15 22:19:32 GMT 1997    MkIII    Start Patrol
COMMENT: Mon Sep 15 22:18:15 GMT 1997
Activity report:
No PAs taken today.
No coronal observations

Tapes:  DPMON:  P00987
        CHIP:   C00408
        LOWL:   L00473

SCAN-LOG
MK3 SCAN-LOG for the above "rawmk3" data, resorted to resemble old format.
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